ORIGINAL RESEARCH

Relationship Between Pain Reduction and Improvement in Health-Related Quality of Life in Patients with Knee Pain Due to Osteoarthritis Receiving Duloxetine: Exploratory Post Hoc Analysis of a Japanese Phase 3 Randomized Study

> This article was published in the following Dove Press journal: Journal of Pain Research

Hiroyuki Enomoto ()¹ Shinji Fujikoshi² Kei Ogawa¹ Toshinaga Tsuji³ Sakae Tanaka ()⁴

 ¹Medicines Development Unit Japan, Eli Lilly Japan K.K., Tokyo 107-0052, Japan;
 ²Medicines Development Unit Japan, Eli Lilly Japan K.K., Kobe 651-0086, Japan;
 ³Medical Affairs Department, Shionogi & Co., Ltd., Shibata, Osaka 530-0012, Japan;
 ⁴Department of Orthopaedic Surgery, Faculty of Medicine, The University of Tokyo, Tokyo, Japan

Correspondence: Hiroyuki Enomoto Medicines Development Unit Japan, Eli Lilly Japan K.K., 4-15-1-13F, Akasaka, Minato-Ku, Tokyo 107-0052, Japan Tel +81 3 5574 9143 Fax +81 3 5574 9979 Email enomoto_hiroyuki@lilly.com



Purpose: This post hoc analysis of a Japanese phase 3 randomized study (ClinicalTrials.gov identifier: NCT02248480) investigated relationships between changes in pain severity and changes in health-related quality of life (HRQoL) in duloxetine-treated patients with knee osteoarthritis (OA).

Patients and Methods: Patients with knee OA and Brief Pain Inventory (BPI) average pain score \geq 4 received duloxetine 60 mg/day or placebo for 14 weeks. Spearman rank correlation coefficients were calculated for change in pain severity, as assessed by the BPI, and change in HRQoL, as assessed by the items of the (i) 36-item Short-Form Health Survey (SF-36; a generic measure of HRQoL) and (ii) Western Ontario and McMaster Universities Osteoarthritis Index (WOMAC; an OA-specific measure of HRQoL).

Results: After 14 weeks of treatment, there was a significantly greater improvement (p<0.001) for duloxetine (n=177) vs placebo (n=176) in BPI average pain severity score and significantly greater improvements (p<0.01) for duloxetine vs placebo for 5 of the 8 SF-36 domains (including the Role-Physical, Bodily Pain, and Physical Functioning domains) and all 24 individual WOMAC items. The correlation between BPI change from baseline and SF-36 item change from baseline was statistically significant (p<0.05) for 2 of the 8 SF-36 items (Bodily Pain, Physical Functioning) in duloxetine-treated patients. The correlation between BPI change from baseline and WOMAC item change from baseline and WOMAC item change from baseline was statistically significant for 22 of the 24 WOMAC items in duloxetine-treated patients.

Conclusion: This post hoc analysis suggested that the pain reduction observed in duloxetine-treated patients with knee OA was associated with improvements in OA-specific aspects of HRQoL, ie, pain and physical functioning.

Keywords: 36-item Short-form Health Survey, Brief Pain Inventory, placebo effect, Western Ontario and McMaster Universities Osteoarthritis Index

Plain Language Summary

• This study looked at people in Japan with pain from knee osteoarthritis (OA). They took duloxetine or a dummy drug (placebo). Overall, people taking duloxetine had bigger improvements in knee pain than people taking placebo.

Journal of Pain Research 2020:13 181–191

© 2020 Enomoto et al. This work is published and licensed by Dove Medical Press Limited. The full terms of this license are available at https://www.dovepress.com/terms. work you hereby accept the Terms. Non-commercial uses of the work are permitted without any further permission form Dove Medical Press Limited, provided the work is properly attributed. For permission for commercial use of this work, lease see paragraphs 4.2 and 5 of our Terms (http://www.dovepress.com/twws.dovepress.com/terms.php).

- The researchers looked at whether people's pain affected their quality of life (QoL) during the study. Everyone rated their pain on a scale of 1 to 10 and completed two QoL questionnaires:
 - o One with general QoL questions.
 - One with questions specifically designed for people with OA.
- Based on answers to the general QoL questions:
 - For the 177 people taking duloxetine, improved pain was related to better QoL for 2 out of 8 question groups. These 2 question groups asked about pain and movement.
 - For the 176 people taking placebo, improved pain was related to better QoL for 7 out of 8 question groups.
- Based on answers to the QoL questions specific to OA:
 - For people taking duloxetine, improved pain was related to better QoL for 22 out of 24 questions.
 - For people taking placebo, improved pain was related to better QoL for all 24 questions.
- These results suggest that for people taking duloxetine, improvements in their pain helped to improve aspects of their QoL related to OA. However, for people taking placebo, improvements in their pain were related to improvements in a broad range of QoL aspects. This could be because they expected the study treatment to be effective and did not know they were taking a placebo (a so-called "placebo effect").

Introduction

Osteoarthritis (OA) of the knee is a disabling chronic condition,^{1,2} which is highly prevalent in Japan. In the large-scale Research on Osteoarthritis Against Disability (ROAD) study conducted in Japan, the prevalence of knee OA in male and female participants aged ≥ 60 years was 47.0% and 70.2%, respectively.³ The primary symptoms of knee OA are joint pain, stiffness, and loss of function, which lead to reduced health-related quality of life (HROoL).^{1,4} Recent publications have also shown an association between painful OA and the risk of cardiovascular disease,^{5,6} which implies that OA pain has wider effects on a person's health and underlines the importance of OA pain management. OA treatment includes surgical and nonsurgical interventions. Nonsurgical interventions comprise nonpharmaceutical and pharmaceutical treatments, with analgesics being one of the key pillars for treating patients with OA. Current pharmaceutical treatments include acetaminophen, oral and topical nonsteroidal anti-inflammatory drugs (NSAIDs), opioids, and intra-articular corticosteroids.^{2,7–9} However, these drugs have limited efficacy, provide benefit for only a proportion of patients, and/or their use is restricted by adverse events such as gastrointestinal bleeding and impairment of renal function.^{2,7,9,10} In particular, opioids are associated with the risk of abuse and addiction, while not showing an advantage over non-opioid medications with respect to improving pain-related function in patients with knee OA.¹¹ In light of the opioid epidemic, other, non-opioid, strategies for pain management in the treatment of knee OA are needed, with the treatment goals of improving HRQoL and physical function as well as reducing pain.

Duloxetine, a selective serotonin-norepinephrine reuptake inhibitor, has shown efficacy in the treatment of knee OA, accompanied by an acceptable safety profile, in several randomized, placebo-controlled studies conducted in the US, Europe, China, and Japan.¹²⁻¹⁶ According to the Osteoarthritis Research Society International (OARSI) treatment guidelines, duloxetine is classified as appropriate for individuals with knee OA without comorbidities and individuals with multiple-joint OA and relevant comorbidities, with uncertain appropriateness for individuals with knee-only OA and comorbidities.⁹ In the randomized, placebo-controlled, phase 3 study conducted in 354 Japanese patients with knee OA,¹⁶ duloxetine significantly reduced pain, as assessed by the Brief Pain Inventory (BPI). In addition, duloxetine improved knee function as assessed by the Ontario and McMaster Western Universities Osteoarthritis Index (WOMAC; an OA-specific measure of pain relief, stiffness, and functional improvement) and improved several aspects of HRQoL as assessed by the 36-item Short-Form Health Survey (SF-36; a generic measure of HRQoL).¹⁶ Reducing pain intensity and improving HROoL are the main treatment goals for knee OA. However, there are limited data on the correlation between improvement in pain severity and improvement in HRQoL in patients with knee OA. Information on which aspects of HRQoL may be improved with pharmaceutical interventions for pain would be useful for physicians in making treatment decisions and informing patients of likely treatment outcomes. In addition, placebo effects have been recognized in clinical trials, especially in the case of analgesics.¹⁷ However, correlations between placebo analgesic effects and improvements in HRQoL are not well known.

The aim of the current post hoc analysis was to investigate correlations in duloxetine-treated patients between change in pain severity, as assessed by the BPI, and change in HRQoL, as assessed by the items of the SF-36 and the WOMAC in the previously conducted phase 3 study of duloxetine vs placebo in Japanese patients with knee OA.¹⁶ In addition, correlations between change in pain severity and change in HRQoL were investigated in placebo-treated patients and compared with the correlations observed in duloxetine-treated patients.

Materials and Methods

Study Design

Full details of the study design have been published elsewhere.¹⁶ This was a multicenter, randomized, placebocontrolled, double-blind, phase 3 study conducted at 47 medical institutions in Japan from October 2014 to June 2015. The study protocol was approved by the Institutional Review Board of each medical institution (<u>Supplementary Table 1</u>) and the study was conducted in accordance with Good Clinical Practice guidelines. All patients provided written informed consent before participating in the study. The study was registered at <u>www.clin</u> <u>icaltrials.gov</u> (NCT02248480).

Study Population

Male and female patients with knee OA were eligible for the study. Additional inclusion criteria were: age 40 to 79 years; pain for ≥ 14 days of each month for 3 months before study entry; BPI average pain score of ≥ 4 ; and satisfying the American College of Rheumatology criteria¹⁸ for idiopathic knee OA (knee pain, bone spurs detected on plain X-ray images, and at least one of age >50 years, morning stiffness resolving within 30 mins, or crepitus). The main exclusion criteria were: previous administration of duloxetine; inflammatory arthritis (eg, rheumatoid arthritis) or autoimmune diseases (except Hashimoto's disease and type 1 diabetes); invasive treatment (eg, joint lavage or intra-articular hyaluronic acid or steroid injections) in either knee within 1 month before Visit 1; arthroscopic surgery of the affected joint within 1 year before Visit 1 or a history of joint replacement or osteotomy; end-stage OA (eg, patients with loss of joint space or loss of articular cartilage) or patients scheduled to undergo surgery of the affected joint during the study; major depressive disorders based on the Mini International Neuropsychiatric Interview.¹⁹

Treatment Protocol

The study period comprised a 1- to 2-week pretreatment period, a 14-week treatment period, a 1-week taper period,

and a 1-week follow-up period. Patients were randomized to receive duloxetine 60 mg once daily (a 20-mg capsule for 1 week, two 20-mg capsules for 1 week, and three 20mg capsules for 12 weeks) or matching placebo tablets. Drugs with analgesic effect (eg, NSAIDs) were permitted as rescue medication for up to 3 consecutive days and for a cumulative total of 20 days.

Assessments

Pain intensity was assessed using the BPI 24-hr average pain severity score (the primary outcome measure of the study¹⁶), which measures average pain during the past 24 hrs.²⁰ The BPI average pain severity score was rated on a scale ranging from 0 ("no pain") to 10 ("pain as bad as you can imagine"). Generic quality of life was assessed using the SF-36, which comprises 36 questions divided into 8 domains: physical functioning, role-physical, bodily pain, general health, vitality, social functioning, roleemotional, and mental health.²¹ Each of the 8 domains was rated on a scale ranging from 0 to 100, with higher scores indicating better health and well-being. Osteoarthritis-specific quality of life was assessed using the WOMAC, which comprises 24 questions divided into three subscales: pain, stiffness, and physical function.²² Each of the WOMAC items was rated on a scale of 0 to 4, with lower scores indicating lower levels of symptoms or physical disability.

Outcome Measures

The efficacy outcome measures in this post hoc analysis were the change from baseline to Week 14 in BPI 24-hr average pain severity score, individual SF-36 item scores, and WOMAC total, pain, stiffness, and physical function subscale, and individual item scores.

Statistical Analysis

This post hoc analysis was performed on the full analysis set, which consisted of all randomized patients who received at least 1 dose of study drug and had a baseline and postbaseline BPI average pain severity score. Analysis of covariance was used to compare the change from baseline to Week 14 for duloxetine vs placebo in BPI average pain severity score, SF-36 item scores, and WOMAC total, subscale, and item scores. Spearman rank correlation coefficients were calculated for (i) change in pain severity (as assessed by the BPI) at Week 14 and change in generic HRQoL (as assessed by the SF-36 items) at Week 14 and (ii) change in pain severity (as assessed by the BPI) at Week 14 and change in OA-specific HRQoL (as assessed by the WOMAC items) at Week 14. The correlation coefficients were calculated separately for duloxetine- and placebo-treated patients. When the value at Week 14 was missing, the last observation carried forward method was used for the imputation. Data were missing for 17/177 patients (9.6%) in the duloxetine group and 15/176 patients (8.5%) in the placebo group for the BPI and each of the items of the SF-36 and WOMAC scales. Statistical tests were performed at a two-sided significance level of 0.05. Analyses were performed using SAS version 9.4 (SAS Institute, Cary, NC, USA).

Results

Demographic and Baseline Clinical Characteristics

In total, 354 patients were randomized, with 353 patients analyzed for efficacy: 177 patients received duloxetine and 176 patients received placebo. The mean (standard deviation [SD]) baseline BPI average score was 5.0 (1.0) and 5.1 (1.0) in the duloxetine and placebo groups, respectively (Table 1). The mean baseline SF-36 item scores were similar between the duloxetine and placebo groups (Table 1). The 2 worstrated individual SF-36 items (lower score = worse status) at baseline in both the duloxetine and placebo groups were SF363 (Bodily Pain; mean [SD]: duloxetine: 45.9 [12.9]; placebo: 45.1 [12.9]) and SF361 (Physical Functioning; mean [SD]: duloxetine: 57.6 [19.1]; placebo: 55.9 [18.2]) (Table 1). The mean baseline WOMAC total scores were similar between the duloxetine and placebo groups, as were the mean baseline WOMAC pain, stiffness, and physical function subscale scores (Table 1). The 2 worst-rated individual WOMAC items (higher score = worse status) at baseline in both the duloxetine and placebo groups were: PAIN2W ("How much pain have you had when going up or down stairs?"; mean [SD]: duloxetine: 2.28 [0.71]; placebo: 2.28 [0.79]) and PFTN8W ("How much difficulty have you had when going down stairs?"; mean [SD]: duloxetine: 2.11 [0.89]; placebo: 2.09 [0.85]) (Table 1).

Effect of Duloxetine Treatment on BPI Average Pain Severity

After 14 weeks of treatment, there was a significantly greater improvement (p<0.001) in the duloxetine group compared with the placebo group in BPI average pain severity score (least squares [LS] mean difference [95% CI]: -0.78 [-1.12, -0.44]).

Effect of Duloxetine Treatment on SF-36 and WOMAC

As reported previously,¹⁶ after 14 weeks of treatment, there were significantly greater improvements (p<0.01) in the duloxetine group compared with the placebo group for 5 of the 8 SF-36 domains (SF361, SF362, SF363, SF364, and SF367; Table 2). The 3 individual SF-36 items with the greatest improvement for duloxetine vs placebo were SF362 (Role-Physical; LS mean difference [95% CI]: 7.78 [4.18, 11.39]), SF363 (Bodily Pain; LS mean difference [95% CI]: 6.69 [3.30, 10.08]), and SF361 (Physical Functioning; LS mean difference [95% CI]: 6.39 [2.85, 9.94]) (Table 2). Two of these 3 items, SF363 and SF361, were the 2 worst-rated SF-36 items at baseline.

As reported previously,¹⁶ after 14 weeks of treatment, there were significantly greater improvements (p<0.05) in the duloxetine group compared with the placebo group in the WOMAC total score and pain, stiffness, and physical subscale scores (Table 2). In addition, there were significantly greater improvements for duloxetine vs placebo in all of the 24 individual WOMAC items (Table 2). The 3 individual WOMAC items with the greatest improvement for duloxetine vs placebo were PAIN4W ("How much pain have you had while sitting or lying down?"; LS mean difference [95% CI]: -0.40 [-0.57, -0.24]), PFTN12W ("How much difficulty have you had when bending to the floor?"; LS mean difference [95% CI]: -0.38 [-0.53, -0.23]), and STIFF6W ("How severe has your stiffness been after you first woke up in the morning?"; LS mean difference [95% CI]: -0.36 [-0.51, -0.20]) (Table 2). The 2 worst-rated WOMAC items at baseline, PAIN2W and PFTN8W, were also significantly improved from baseline.

Correlations Between BPI Change from Baseline and SF-36 Item Change from Baseline

In duloxetine-treated patients, the correlation between BPI change from baseline and SF-36 item change from baseline was statistically significant (p<0.05) for 2 of the 8 SF-36 items (25%; Table 3). The correlation coefficient was negative for all SF-36 items, meaning that a decrease in BPI score (ie, improvement in pain severity) was associated with an increase in SF-36 item score (ie, improvement in HRQoL). The two strongest correlations between BPI change from baseline and SF-36 item change from baseline were SF363 (Bodily Pain; Spearman correlation:

Table I Patient Demographics and Baseline Characteristics

Parameter ^a	Duloxetine	Placebo	Total
	n=177	n=176	n=353
Age, years	65.5 (8.0)	66.4 (8.4)	65.9 (8.2)
Male, n (%)	35 (19.8)	44 (25.0)	79 (22.4)
Duration of OA, years	4.0 (4.2)	4.5 (4.3)	4.2 (4.2)
BPI average pain score	5.0 (1.0)	5.1 (1.0)	5.0 (1.0)
SF-36 score			
SF361: Physical Functioning	57.6 (19.1)	55.9 (18.2)	56.8 (18.7)
SF362: Role-Physical ^b	71.2 (23.7)	73.5 (20.0)	72.3 (21.9)
SF363: Bodily Pain	45.9 (12.9)	45.1 (12.9)	45.5 (12.9)
SF364: General Health	62.7 (16.0)	64.0 (16.3)	63.4 (16.1)
SF365: Vitality	66.3 (18.1)	65.4 (16.8)	65.9 (17.4)
SF366: Social Functioning	84.7 (19.3)	85.8 (18.9)	85.2 (19.1)
SF367: Role-Emotional ^c	80.8 (23.5)	85.8 (17.4)	83.3 (20.8)
SF368: Mental Health	79.0 (17.0)	78.4 (14.8)	78.7 (15.9)
WOMAC ^d total score	32.67 (13.18)	32.70 (13.71)	32.68 (13.43
WOMAC pain subscale score	7.46 (2.66)	7.55 (2.74)	7.50 (2.70)
PAINIW: Walking on a flat surface	1.41 (0.74)	1.45 (0.70)	1.43 (0.72)
PAIN2W: Going up or down stairs	2.28 (0.71)	2.28 (0.79)	2.28 (0.75)
PAIN3W: At night while in bed (pain that disturbs your sleep)	0.98 (0.85)	0.97 (0.80)	0.98 (0.83)
PAIN4W: While sitting or lying down	1.62 (0.83)	1.72 (0.89)	1.67 (0.86)
PAIN5W: While standing	1.16 (0.74)	1.12 (0.86)	1.14 (0.80)
WOMAC stiffness subscale score	3.01 (1.49)	3.02 (1.55)	3.01 (1.52)
STIFF6W: On first waking up in the morning	1.39 (0.91)	1.41 (0.90)	1.40 (0.91)
STIFF7W: After sitting or lying down or while resting later in the day	1.62 (0.80)	1.60 (0.87)	1.61 (0.84)
WOMAC physical function subscale score	22.20 (10.21)	22.14 (10.51)	22.17 (10.3
PFTN8W: Difficulty going down stairs	2.11 (0.89)	2.09 (0.85)	2.10 (0.87)
PFTN9W: Difficulty going up stairs	1.73 (0.82)	1.68 (0.86)	1.71 (0.83)
PFTN10W: Difficulty getting up from a sitting position	1.90 (0.94)	1.92 (0.94)	1.91 (0.94)
PFTN11W: Difficulty while standing	1.03 (0.77)	1.05 (0.82)	1.04 (0.79)
PFTN12W: Difficulty bending to the floor	1.53 (0.85)	1.52 (0.89)	1.52 (0.87)
PFTN13W: Difficulty walking on a flat surface	1.18 (0.81)	1.11 (0.77)	1.14 (0.79)
PFTN14W: Difficulty getting in or out of a car, or getting on or off a bus	1.35 (0.84)	1.38 (0.89)	1.36 (0.87)
PFTN15W: Difficulty while going shopping	1.17 (0.83)	1.19 (0.78)	1.18 (0.80)
PFTN16W: Difficulty putting on socks or panty hose or stockings	1.11 (0.84)	1.16 (0.85)	1.14 (0.84)
PFTN17W: Difficulty getting out of bed	1.14 (0.88)	1.17 (0.84)	1.16 (0.86)
PFTN18W: Difficulty taking off socks or panty hose or stockings	1.11 (0.85)	1.09 (0.85)	1.10 (0.85)
PFTN19W: Difficulty while lying in bed	0.66 (0.74)	0.72 (0.74)	0.69 (0.74)
PFTN20W: Difficulty getting in or out of the bathtub	1.06 (0.86)	0.98 (0.81)	1.02 (0.84)
PFTN21W: Difficulty while sitting	1.64 (1.09)	1.63 (1.04)	1.63 (1.07)
PFTN22W: Difficulty getting on or off the toilet	0.91 (0.84)	0.88 (0.89)	0.89 (0.86)
PFTN23W: Difficulty doing heavy household chores	1.80 (0.87)	1.81 (0.93)	1.80 (0.90)

Notes: ^aExcept where otherwise indicated, data are mean (standard deviation). ^bRole limitations because of physical problems. ^cRole limitations because of emotional problems. ^dWOMAC scale items from Bellamy N, Buchanan WW, Goldsmith CH, et al. Validation study of WOMAC: a health status instrument for measuring clinically important patient relevant outcomes to antirheumatic drug therapy in patients with osteoarthritis of the hip or knee. *J Rheumatol.* 1988;15(12):1833–1840. Copyright © 1996 Nicholas Bellamy. All Rights Reserved.²⁸

Abbreviations: BPI, Brief Pain Inventory; OA, osteoarthritis; SF-36, 36-item Short Form Survey; WOMAC, Western Ontario and McMaster Universities Osteoarthritis Index.

Table 2 Least Squares Mean Change from Baseline to Week 14 in SF-36 Subscale Scores and WOMAC Total and Subscale Scores in
Duloxetine- and Placebo-Treated Patients

Scale Item ^a	Duloxetine n=177 LS Mean (SE)	Placebo n=176 LS Mean (SE)	LS Mean Difference (95% CI)	P-value
SF-36 score				
SF361	12.62 (1.27)	6.23 (1.27)	6.39 (2.85, 9.94)	<0.001
SF362	11.44 (1.29)	3.66 (1.30)	7.78 (4.18, 11.39)	<0.001
SF363	16.32 (1.22)	9.63 (1.22)	6.69 (3.30, 10.08)	<0.001
SF364	5.58 (0.93)	1.82 (0.94)	3.77 (1.16, 6.37)	0.005
SF365	3.99 (1.04)	3.16 (1.04)	0.83 (-2.05, 3.72)	0.570
SF366	5.66 (1.14)	2.54 (1.15)	3.12 (-0.07, 6.31)	0.055
SF367	6.32 (1.27)	0.70 (1.28)	5.62 (2.06, 9.17)	0.002
SF368	3.02 (0.99)	1.48 (0.99)	1.55 (-1.21, 4.30)	0.270
WOMAC total score	-16.94 (0.90)	-9.99 (0.90)	-6.95 (-9.47, -4.44)	<0.001
WOMAC pain subscale score	-3.90 (0.21)	-2.31 (0.21)	-1.59 (-2.17, -1.01)	<0.001
PAINTW	-0.81 (0.05)	-0.48 (0.05)	-0.33 (-0.47, -0.19)	<0.001
PAIN2W	-1.01 (0.06)	-0.71 (0.06)	-0.30 (-0.46, -0.13)	<0.001
PAIN3W	-0.63 (0.05)	-0.34 (0.05)	-0.29 (-0.42, -0.16)	<0.001
PAIN4W	-0.79 (0.06)	-0.39 (0.06)	-0.40 (-0.57, -0.24)	<0.001
PAIN5W	-0.62 (0.05)	-0.34 (0.05)	-0.28 (-0.41, -0.15)	<0.001
WOMAC stiffness subscale score	-1.61 (0.09)	-0.91 (0.09)	-0.70 (-0.96, -0.44)	<0.001
STIFF6W	-0.74 (0.05)	-0.38 (0.06)	-0.36 (-0.51, -0.20)	<0.001
STIFF7W	-0.83 (0.05)	-0.53 (0.05)	-0.30 (-0.44, -0.16)	<0.001
WOMAC physical function subscale score	-11.44 (0.65)	-6.76 (0.66)	-4.67 (-6.50, -2.85)	<0.001
PFTN8W	-0.84 (0.06)	-0.58 (0.06)	-0.26 (-0.44, -0.09)	0.004
PFTN9W	-0.74 (0.06)	-0.45 (0.06)	-0.29 (-0.45, -0.13)	<0.001
PFTNIOW	-0.93 (0.06)	-0.60 (0.06)	-0.33 (-0.50, -0.16)	<0.001
PFTNIIW	-0.62 (0.05)	-0.33 (0.05)	-0.28 (-0.41, -0.15)	<0.001
PFTN12W	-0.83 (0.05)	-0.45 (0.05)	-0.38 (-0.53, -0.23)	<0.001
PFTN13W	-0.65 (0.05)	-0.41 (0.05)	-0.24 (-0.37, -0.11)	<0.001
PFTN14W	-0.70 (0.05)	-0.36 (0.05)	-0.34 (-0.48, -0.20)	<0.001
PFTN15W	-0.62 (0.05)	-0.41 (0.05)	-0.21 (-0.34, -0.07)	0.003
PFTN16W	-0.60 (0.05)	-0.35 (0.05)	-0.25 (-0.39, -0.11)	<0.001
PFTN17W	-0.73 (0.04)	-0.42 (0.04)	-0.32 (-0.44, -0.19)	<0.001
PFTN18W	-0.56 (0.05)	-0.32 (0.05)	-0.25 (-0.39, -0.11)	<0.001
PFTN19W	-0.44 (0.04)	-0.30 (0.04)	-0.14 (-0.25, -0.03)	0.014
PFTN20W	-0.53 (0.05)	-0.26 (0.05)	-0.27 (-0.39, -0.14)	<0.001
PFTN21W	-0.72 (0.06)	-0.49 (0.06)	-0.23 (-0.41, -0.05)	0.012
PFTN22W	-0.56 (0.04)	-0.35 (0.04)	-0.21 (-0.33, -0.09)	<0.001
PFTN23W	-0.72 (0.06)	-0.47 (0.06)	-0.25 (-0.42, -0.08)	0.005
PFTN24W	-0.47 (0.04)	-0.22 (0.04)	-0.25 (-0.37, -0.13)	<0.001

Notes: ^aSee Table I for a description of each item. WOMAC scale items from Bellamy N, Buchanan WW, Goldsmith CH, et al. Validation study of WOMAC: a health status instrument for measuring clinically important patient relevant outcomes to antirheumatic drug therapy in patients with osteoarthritis of the hip or knee. *J Rheumatol.* 1988;15(12):1833–1840. Copyright © 1996 Nicholas Bellamy. All Rights Reserved.²⁸

Abbreviations: Cl, confidence interval; LS, least squares; SE, standard error; SF-36, 36-item Short Form Survey; WOMAC, Western Ontario and McMaster Universities Osteoarthritis Index.

-0.452) and SF361 (Physical Functioning; Spearman correlation: -0.289) (Table 3).

from baseline in placebo-treated patients was statistically significant (p<0.05) for 7 of the 8 SF-36 items (87.5%; Table 3). As observed for duloxetine-treated patients, the two strongest correlations between BPI change from baseline

In contrast to duloxetine-treated patients, the correlation between BPI change from baseline and SF-36 item change

Scale Item ^a	Duloxetine n=177	Duloxetine n=177		Placebo n=176	
	Spearman Correlation	P-value	Spearman Correlation	P-value	
SF363: Bodily Pain	-0.452	<0.001	-0.502	<0.001	
SF361: Physical Functioning	-0.289	<0.001	-0.326	<0.001	
SF365: Vitality	-0.135	0.073	-0.261	<0.001	
SF368: Mental Health	-0.130	0.084	-0.225	0.003	
SF362: Role-Physical ^b	-0.089	0.241	-0.285	<0.001	
SF366: Social Functioning	-0.079	0.298	-0.136	0.072	
SF364: General Health	-0.055	0.470	-0.244	0.001	
SF367: Role-Emotional ^c	-0.011	0.884	-0.164	0.030	

 Table 3 Correlation Between BPI Change from Baseline and SF-36 Change from Baseline in Duloxetine-Treated and Placebo-Treated

 Patients

Notes: ^aIndividual SF-36 items arranged in descending order of the size of Spearman correlation coefficient in duloxetine-treated patients. ^bRole limitations because of physical problems. ^cRole limitations because of emotional problems.

Abbreviations: BPI, Brief Pain Inventory; SF-36, 36-item Short-Form Survey.

and SF-36 item change from baseline for placebo-treated patients were SF363 (Bodily Pain; Spearman correlation: -0.502) and SF361 (Physical Functioning: -0.289; placebo: -0.326) (Table 3). For each of the SF-36 items, the size of the correlation coefficient was larger in placebo-treated patients than duloxetine-treated patients.

Correlations Between BPI Change from Baseline and WOMAC Item Change from Baseline

In duloxetine-treated patients, the correlation between BPI change from baseline and WOMAC item change from baseline was statistically significant (p<0.05) for the total score, the pain, physical function, and stiffness subscale scores, and 22 of the 24 WOMAC items (91.7%; Table 4). The strongest correlations between BPI change from baseline and WOMAC item change from baseline were PFTN23W ("How much difficulty have you had while doing heavy household chores?"; Spearman correlation: 0.428), PAIN2W ("How much pain have you had when going up or down stairs?"; Spearman correlation: 0.421), and PFTN24W ("How much difficulty have you had while doing light household chores?"; Spearman correlation: 0.401) (Table 4).

In placebo-treated patients, the correlation between BPI change from baseline and WOMAC item change from baseline was statistically significant (p<0.001) for the total score, the pain, physical function, and stiffness subscale scores, and all 24 WOMAC items (100%; Table 4). The strongest correlations between BPI change from baseline and WOMAC item change from baseline were PAIN2W ("How much pain have you had when going up or down stairs?"; Spearman correlation: 0.591), PFTN8W ("How much difficulty have you had

when going down stairs?"; Spearman correlation: 0.555), and PAIN1W ("How much pain have you had when walking on a flat surface?"; Spearman correlation: 0.539) (Table 4). For most of the WOMAC items, the size of the correlation coefficient was larger in placebo-treated patients than in duloxetine-treated patients.

Discussion

Treatment goals for patients with knee OA include improving HRQoL as well as reducing pain intensity. Few studies have demonstrated correlations between pain reduction attributed to analgesics with the amelioration of HROoL. The primary report from this study¹⁶ showed that duloxetine treatment both reduced pain (as assessed by the BPI) and improved HRQoL (as assessed by the SF-36 and WOMAC) in Japanese patients with knee OA. This post hoc analysis extends these findings by showing that there were correlations between pain reduction in duloxetinetreated patients and improvements in the SF-36 items related to pain and physical functioning and improvements in nearly all items of the OA-specific WOMAC. These results suggest that the pain reduction resulting from duloxetine treatment is associated with improvements in the disease-specific aspects of HRQoL that are compromised in patients with knee OA.

In duloxetine-treated patients, the correlation between pain reduction and SF-36 item change from baseline was statistically significant for 2 of the 8 SF-36 items, namely, SF363 (Bodily Pain) and SF361 (Physical Functioning). In contrast, in placebo-treated patients, the correlation between pain reduction and SF-36 item change from baseline was statistically significant for all SF-36 items except SF366

187

Scale Item ^a	Duloxetine n=177		Placebo n=176	
	Spearman Correlation	P-value	Spearman Correlation	P-value
WOMAC total score	0.420	<0.001	0.577	<0.001
WOMAC pain subscale score	0.446	<0.001	0.596	<0.001
WOMAC physical function subscale score	0.389	<0.001	0.543	<0.001
WOMAC stiffness subscale score	0.230	0.002	0.325	<0.001
WOMAC items ^b				
PFTN23W	0.428	<0.001	0.391	<0.001
PAIN2W	0.421	<0.001	0.591	<0.001
PFTN24W	0.401	<0.001	0.308	<0.001
PAINIW	0.386	<0.001	0.539	<0.001
PFTN14W	0.366	<0.001	0.352	<0.001
PFTN8W	0.333	<0.001	0.555	<0.001
PFTN15W	0.312	<0.001	0.438	<0.001
PFTN16W	0.309	<0.001	0.348	<0.001
PFTN12W	0.306	<0.001	0.416	<0.001
PAIN5W	0.294	<0.001	0.350	<0.001
PFTN22W	0.278	<0.001	0.292	<0.001
PFTN9W	0.278	<0.001	0.462	<0.001
PAIN3W	0.260	<0.001	0.360	<0.001
PFTN13W	0.252	<0.001	0.415	<0.001
PAIN4W	0.239	0.001	0.375	<0.001
PFTN18W	0.237	0.001	0.330	<0.001
PFTN17W	0.234	0.002	0.340	<0.001
PFTNIIW	0.212	0.005	0.381	<0.001
STIFF6W	0.194	0.010	0.350	<0.001
PFTNI0W	0.177	0.018	0.391	<0.001
STIFF7W	0.172	0.022	0.256	<0.001
PFTN20W	0.171	0.023	0.363	<0.001
PFTN21W	0.134	0.076	0.350	<0.001
PFTN19W	0.119	0.115	0.342	<0.001

 Table 4 Correlation Between BPI Change from Baseline and WOMAC Change from Baseline in Duloxetine-Treated and Placebo-Treated Patients

Notes: ^aSee Table I for a description of each item. WOMAC scale items from Bellamy N, Buchanan WW, Goldsmith CH, et al. Validation study of WOMAC: a health status instrument for measuring clinically important patient relevant outcomes to antirheumatic drug therapy in patients with osteoarthritis of the hip or knee. *J Rheumatol.* 1988;15(12):1833–1840. Copyright © 1996 Nicholas Bellamy. All Rights Reserved.^{28 b}Individual WOMAC items arranged in descending order of the size of Spearman correlation coefficient in duloxetine-treated patients.

Abbreviations: BPI, Brief Pain Inventory; WOMAC, Western Ontario and McMaster Universities Osteoarthritis Index.

(Social Functioning). These results suggest that pain reduction in response to duloxetine treatment was associated with improvements in specific aspects of HRQoL related to OA (ie, pain and physical functioning, which were the 2 worstrated SF-36 items at baseline), while the pain reduction in response to placebo treatment appeared to be associated with broader effects on HRQoL, with improvements in general aspects of HRQoL as well as pain and physical functioning. Intriguingly, the correlation coefficient values were higher in placebo-treated patients than in duloxetine-treated patients. However, it should be noted that the magnitude of improvement in the SF-36 items was smaller in placebo-treated patients than in duloxetine-treated patients, which may be suggestive of a placebo effect. It has been reported that expectation of benefit is one of the mechanisms contributing to placebo analgesia, which activates the reward system of the brain to reduce pain.^{23–25} It might be possible that different mechanisms of action between duloxetine and placebo in reducing pain are responsible for the different patterns of correlations between pain reduction and improvement in HRQoL, as assessed by the generic HRQoL measure SF-36, in the two treatment groups.

Duloxetine treatment significantly improved all of the individual WOMAC items in this study population compared

with placebo, suggesting that pain reduction from duloxetine treatment was associated with improvement in OA-specific aspects of HRQoL in patients with knee OA. This is in contrast to the SF-36 results, where 5 of the 8 items were significantly improved following duloxetine treatment. In addition, a higher proportion of WOMAC items (22 of 24) than SF-36 items (2 of 8) showed statistically significant correlations between pain reduction and improvement in HRQoL item in duloxetine-treated patients. These findings are possibly related to the SF-36 being a generic HRQoL measure that evaluates multiple aspects of HRQoL, not just those aspects specifically associated with OA. In particular, patients with knee OA are likely to have comorbid conditions contributing to their overall HRQoL; the aspects of HRQoL related to such conditions may be captured by the SF-36 items not improved by duloxetine treatment.^{26,27}

In duloxetine-treated patients, the correlations between pain reduction and WOMAC item change from baseline were statistically significant for all but 2 of the 24 items, indicating that pain reduction from duloxetine treatment was associated with improvement in aspects of HRQoL associated with knee OA. The two WOMAC items for which the correlations were not statistically significant (PFTN21W: How much difficulty have you had while sitting?; PFTN19W: How much difficulty have you had while lying in bed?) might be less sensitive indicators of the pain reduction associated with duloxetine treatment. In placebo-treated patients, the correlations between pain reduction and WOMAC item change from baseline were statistically significant for all 24 WOMAC items. As observed for the SF-36, the magnitude of improvement in the WOMAC items was smaller in placebotreated patients than duloxetine-treated patients, and the correlation coefficient values were higher in placebo-treated patients than duloxetine-treated patients, which may be suggestive of a placebo effect. However, we note the difficulty of determining a placebo effect based on high correlation coefficient values given that correlation coefficient values can be high even if both pain and HRQoL do not improve. Placebo effects on OA-specific aspects of HRQoL have been documented, with a meta-analysis of 198 trials with 193 placebo groups (16,364 patients) showing that placebo treatment was effective in relieving pain (usually assessed by the WOMAC pain score) and improving stiffness and function in patients with hand, hip, or knee OA.¹⁷ It may be noted that the three WOMAC items for which the correlation between pain reduction and improvement in HRQoL was stronger in duloxetinetreated patients than placebo-treated patients were related to physical function, in particular, doing heavy (PFTN23W) and

light (PFTN24W) household chores and getting in or out of a car (PFTN14W).

Strengths of this analysis include the study being a randomized, placebo-controlled, double-blind, phase 3 study and the use of individual item scores in the HRQoL analyses. The use of two HRQoL measures allowed assessment of the effect of duloxetine treatment on different aspects of HRQoL. The SF-36, being a generic measure, assesses the effect of treatment on a patient's general health, including assessment of side effects of treatment that may be unrelated to the condition. In contrast, the WOMAC is an OA-specific measure, which assesses the effect of treatment on the domains of HRQoL directly affected by the condition itself. In addition to its post hoc nature, limitations of this analysis include the study being conducted in Japanese patients only and the treatment duration being relatively short and, therefore, the analysis results might not be applicable for other populations or longer treatment durations. Another limitation of the correlation analyses is that it is not possible to determine a causal direction for the correlations, only to state that there was an association between pain reduction and improvement in HRQoL. More complex analyses, for example, mediation models, are required to determine the direction of the relationship between these two parameters. We note that there may be other interactions that influence the associations between pain reduction and improvement in HRQoL. For example, if individual HRQoL items are correlated with each other, then it is difficult to assess whether the correlation between the change in HRQoL item and the change in pain item is a direct or indirect effect. We also note the exploratory nature of these analyses and that the results should be interpreted with caution. Future research into the relationship between pain and HRQoL in patients with knee pain from OA requires studies specifically designed to examine this relationship. Such studies should employ statistical methods capable of interpreting complex interactions between pain and HRQoL, as assessed by the instruments constructed to evaluate these parameters.

In conclusion, this post hoc analysis found different correlation profiles for pain reduction and improvements in HRQoL between duloxetine- and placebo-treated patients with knee OA. The results suggest that the pain reduction observed in this population of duloxetine-treated patients was associated with improvements in OA-specific aspects of HRQoL. In contrast, the pain reduction observed in placebo-treated patients was associated with improvements in a broad spectrum of HRQoL domains, possibly due to a placebo effect.

Data Sharing Statement

Individual participant data have not been made available at this time; however, future requests for de-identified patient-level data may be considered.

Acknowledgment

The authors would like to thank all study participants.

Author Contributions

All authors participated in the interpretation of study results, and in the drafting, critical revision, and approval of the final version of the manuscript, and all authors agree to be accountable for all aspects of the work. HE was involved in the design of the analyses and SF was involved in the statistical analyses.

Funding

This study was sponsored by Shionogi & Co., Ltd., Eli Lilly Japan K.K., and Eli Lilly and Company. Medical writing assistance was provided by Justine Southby, PhD, CMPP, and Tania Dickson, PhD, CMPP, of ProScribe – Envision Pharma Group, and was funded by Eli Lilly Japan K.K. ProScribe's services complied with international guidelines for Good Publication Practice (GPP3).

Role of the Sponsor

Shionogi & Co., Ltd., Eli Lilly Japan K.K., and Eli Lilly and Company were involved in the study design, data collection, data analysis, and preparation of the manuscript.

Disclosure

TT is an employee of Shionogi & Co., Ltd. and owns stock in Shionogi & Co., Ltd. and Takeda Pharmaceutical Company Ltd. HE, SF, and KO are employees of Eli Lilly Japan K.K. HE, SF, and KO own stock in Eli Lilly and Company. ST has received research grants from Eli Lilly Japan K.K. The authors report no other conflicts of interest in this work.

References

- Neogi T. The epidemiology and impact of pain in osteoarthritis. Osteoarthritis Cartilage. 2013;21(9):1145–1153. doi:10.1016/j. joca.2013.03.018
- Mora JC, Przkora R, Cruz-Almeida Y. Knee osteoarthritis: pathophysiology and current treatment modalities. *J Pain Res.* 2018;11:2189–2196. doi:10.2147/JPR
- Muraki S, Oka H, Akune T, et al. Prevalence of radiographic knee osteoarthritis and its association with knee pain in the elderly of Japanese population-based cohorts: the ROAD study. *Osteoarthritis Cartilage*. 2009;17(9):1137–1143. doi:10.1016/j.joca.2009.04.005

- 4. Brown JP, Boulay LJ. Clinical experience with duloxetine in the management of chronic musculoskeletal pain. A focus on osteoarthritis of the knee. *Ther Adv Musculoskelet Dis*. 2013;5(6):291–304. doi:10.1177/1759720X13508508
- 5. Rahman MM, Kopec JA, Cibere J, Goldsmith CH, Anis AH. The relationship between osteoarthritis and cardiovascular disease in a population health survey: a cross-sectional study. *BMJ Open*. 2013;3(5). doi:10.1136/bmjopen-2013-002624
- Wang H, Bai J, He B, Hu X, Liu D. Osteoarthritis and the risk of cardiovascular disease: a meta-analysis of observational studies. *Sci Rep.* 2016;6:39672. doi:10.1038/srep39672
- Hochberg MC, Altman RD, April KT, et al. American College of Rheumatology 2012 recommendations for the use of nonpharmacologic and pharmacologic therapies in osteoarthritis of the hand, hip, and knee. *Arthritis Care Res.* 2012;64(4):465–474. doi:10.1002/ acr.21596
- Jevsevar DS. Treatment of osteoarthritis of the knee: evidence-based guideline, 2nd edition. J Am Acad Orthop Surg. 2013;21(9):571–576. doi:10.5435/JAAOS-21-09-571
- McAlindon TE, Bannuru RR, Sullivan MC, et al. OARSI guidelines for the non-surgical management of knee osteoarthritis. *Osteoarthritis Cartilage*. 2014;22(3):363–388. doi:10.1016/j.joca. 2014.01.003
- Walsh DA, Stocks J. New therapeutic targets for osteoarthritis pain. SLAS Discov. 2017;22(8):931–949. doi:10.1177/2472555217716912
- 11. Krebs EE, Gravely A, Nugent S, et al. Effect of opioid vs nonopioid medications on pain-related function in patients with chronic back pain or hip or knee osteoarthritis pain: the SPACE randomized clinical trial. JAMA. 2018;319(9):872–882. doi:10.1001/jama. 2018.0899
- Chappell AS, Ossanna MJ, Liu-Seifert H, et al. Duloxetine, a centrally acting analgesic, in the treatment of patients with osteoarthritis knee pain: a 13-week, randomized, placebo-controlled trial. *Pain.* 2009;146(3):253–260. doi:10.1016/j.pain.2009.06.024
- 13. Chappell AS, Desaiah D, Liu-Seifert H, et al. A double-blind, randomized, placebo-controlled study of the efficacy and safety of duloxetine for the treatment of chronic pain due to osteoarthritis of the knee. *Pain Pract.* 2011;11(1):33–41. doi:10.1111/ppr.2011.11.issue-1
- 14. Frakes EP, Risser RC, Ball TD, Hochberg MC, Wohlreich MM. Duloxetine added to oral nonsteroidal anti-inflammatory drugs for treatment of knee pain due to osteoarthritis: results of a randomized, double-blind, placebo-controlled trial. *Curr Med Res Opin.* 2011;27 (12):2361–2372. doi:10.1185/03007995.2011.633502
- Wang G, Bi L, Li X, et al. Efficacy and safety of duloxetine in Chinese patients with chronic pain due to osteoarthritis: a randomized, double-blind, placebo-controlled study. *Osteoarthritis Cartilage*. 2017;25(6):832–838. doi:10.1016/j.joca.2016.12.025
- 16. Uchio Y, Enomoto H, Alev L, et al. A randomized, double-blind, placebo-controlled Phase III trial of duloxetine in Japanese patients with knee pain due to osteoarthritis. *J Pain Res.* 2018;11:809–821. doi:10.2147/JPR
- 17. Zhang W, Robertson J, Jones AC, Dieppe PA, Doherty M. The placebo effect and its determinants in osteoarthritis: meta-analysis of randomised controlled trials. *Ann Rheum Dis.* 2008;67 (12):1716–1723. doi:10.1136/ard.2008.092015
- Altman R, Asch E, Bloch D, et al. Development of criteria for the classification and reporting of osteoarthritis. Classification of osteoarthritis of the knee. Diagnostic and therapeutic criteria committee of the American Rheumatism Association. *Arthritis Rheum*. 1986;29 (8):1039–1049. doi:10.1002/art.1780290816
- Sheehan D, Lecrubier Y. M.I.N.I.: International Neuropsychiatric Interview. Japanese 5.0.0 Ed. Tokyo: Seiwa Shoten Co. Ltd; 2003.
- 20. Uki J, Mendoza T, Cleeland CS, Nakamura Y, Takeda F. A brief cancer pain assessment tool in Japanese: the utility of the Japanese Brief Pain Inventory–BPI-J. *J Pain Symptom Manage*. 1998;16 (6):364–373. doi:10.1016/S0885-3924(98)00098-0

- Fukuhara S, Bito S, Green J, Hsiao A, Kurokawa K. Translation, adaptation, and validation of the SF-36 Health Survey for use in Japan. J Clin Epidemiol. 1998;51(11):1037–1044. doi:10.1016/S0895-4356(98)00095-X
- 22. Hashimoto H, Hanyu T, Sledge CB, Lingard EA. Validation of a Japanese patient-derived outcome scale for assessing total knee arthroplasty: comparison with Western Ontario and McMaster Universities osteoarthritis index (WOMAC). J Orthop Sci. 2003;8 (3):288–293. doi:10.1007/s10776-002-0629-0
- Scott DJ, Stohler CS, Egnatuk CM, Wang H, Koeppe RA, Zubieta JK. Individual differences in reward responding explain placebo-induced expectations and effects. *Neuron*. 2007;55 (2):325–336. doi:10.1016/j.neuron.2007.06.028
- Navratilova E, Porreca F. Reward and motivation in pain and pain relief. Nat Neurosci. 2014;17(10):1304–1312. doi:10.1038/nn.3811

- Medoff ZM, Colloca L. Placebo analgesia: understanding the mechanisms. *Pain Manag.* 2015;5(2):89–96. doi:10.2217/pmt.15.3
- 26. Tuominen U, Blom M, Hirvonen J, et al. The effect of co-morbidities on health-related quality of life in patients placed on the waiting list for total joint replacement. *Health Qual Life Outcomes*. 2007;5:16. doi:10.1186/1477-7525-5-16
- 27. van Dijk GM, Veenhof C, Schellevis F, et al. Comorbidity, limitations in activities and pain in patients with osteoarthritis of the hip or knee. *BMC Musculoskelet Disord*. 2008;9:95. doi:10.1186/1471-2474-9-95
- 28. Bellamy N, Buchanan WW, Goldsmith CH, Campbell J, Stitt LW. Validation study of WOMAC: a health status instrument for measuring clinically important patient relevant outcomes to antirheumatic drug therapy in patients with osteoarthritis of the hip or knee. *J Rheumatol.* 1988;15(12):1833–1840.

Journal of Pain Research

Dovepress

Publish your work in this journal

The Journal of Pain Research is an international, peer reviewed, open access, online journal that welcomes laboratory and clinical findings in the fields of pain research and the prevention and management of pain. Original research, reviews, symposium reports, hypothesis formation and commentaries are all considered for publication. The manuscript

management system is completely online and includes a very quick and fair peer-review system, which is all easy to use. Visit http:// www.dovepress.com/testimonials.php to read real quotes from published authors.

Submit your manuscript here: https://www.dovepress.com/journal-of-pain-research-journal